Author’s response to reviews

Title: Adenocarcinoma of the bladder following nephrogenic adenoma: a Case report.

Authors:

   Edwin Hungerhuber (Hungerhuber@gmx.de)
   Ekkehard Bach (Dr.Ekkehard.Bach@t-online.de)
   Arndt Hartmann (arndt.hartmann@klinik.uni-regensburg.de)
   Christian G Stief (Christian.Stief@uro.med.uni-muenchen.de)
   Dirk Zaak (Dirk.Zaak@uro.med.uni-muenchen.de)

Version: 2 Date: 2 October 2007

Author’s response to reviews:

A) the subheading "Background" has been changed to "Introduction"
B) 3 figures were removed
C) the manuscript has been revised